Author's response to reviews

Title: SCN5A allelic expression imbalance in African-Americans heterozygous for the common variant p.Ser1103Tyr

Authors:

Stacy Killen (stacy.stratemann@Vanderbilt.Edu)
Jennifer Kunic (jennifer.babcock@Vanderbilt.Edu)
Lily Wang (lily.wang@vanderbilt.edu)
Adele Lewis (alewis@forensicmed.com)
Bruce Levy (BLEVY@forensicmed.com)
Michael Ackerman (ackerman.michael@mayo.edu)
Alfred George (al.george@vanderbilt.edu)

Version: 3 Date: 10 May 2010

Author's response to reviews: see over
Title Page

1. Title changed to sentence case. There is one period in the title that is unavoidable without breaching the standard notation a nonsynonymous variant (i.e., p.Ser1103Tyr).

2. Authors’ names and affiliations were modified as requested

3. Updated email address for one author (Levy)

Manuscript body

1. Highlighting removed

2. Authors’ contributions modified as requested

3. Acknowledgments updated to include new grant support

4. Figure references are all consistently “Fig.”

Figure

Revised as requested